Author’s response to reviews

Title: Obesity, metabolic syndrome, impaired fasting glucose and microvascular dysfunction: a principal component analysis approach

Authors:

Diogo G Panazzolo (guarnipana@yahoo.com.br)
Luiz G Kraemer-Aguiar (gkraemer@ig.com.br)
Ruth Clapauch (rclapauch@uol.com.br)
Priscila A Maranhão (priscilamaranhao@gmail.com)
Fernando Sicuro (fsicuro@gmail.com)
Eliete Bouskela (eliete_bouskela@yahoo.com.br)

Version: 2 Date: 10 January 2012

Author’s response to reviews:

Rio de Janeiro, January 09th, 2012

Miss Marinette Lacson
Journal Editorial Office
BioMed Central

Dear Marinette Lacson

Requested changes were made in the manuscript (consent statement and approval of the ethics committee).

Thanks for your consideration,

L. G. Kraemer-Aguiar
Associate professor
Endocrinology – Internal Medicine Department
State University of Rio de Janeiro